

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Schmöller, Michaela; Spitzer, Martin

# Working Paper Endogenous TFP, business cycle persistence and the productivity slowdown

Bank of Finland Research Discussion Papers, No. 21/2019

### **Provided in Cooperation with:** Bank of Finland, Helsinki

*Suggested Citation:* Schmöller, Michaela; Spitzer, Martin (2019) : Endogenous TFP, business cycle persistence and the productivity slowdown, Bank of Finland Research Discussion Papers, No. 21/2019, ISBN 978-952-323-296-9, Bank of Finland, Helsinki, https://nbn-resolving.de/urn:nbn:fi:bof-201909181490

This Version is available at: https://hdl.handle.net/10419/212455

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU

Bank of Finland Research Discussion Papers 21 • 2019

Michaela Schmöller – Martin Spitzer

Endogenous TFP, business cycle persistence and the productivity slowdown



Bank of Finland Research Bank of Finland Research Discussion Papers Editor-in-Chief Esa Jokivuolle

Bank of Finland Research Discussion Paper 21/2019 18 September 2019

Michaela Schmöller – Martin Spitzer Endogenous TFP, business cycle persistence and the productivity slowdown

ISBN 978-952-323-296-9, online ISSN 1456-6184, online

Bank of Finland Research Unit

PO Box 160 FIN-00101 Helsinki

Phone: +358 9 1831

Email: research@bof.fi Website: www.suomenpankki.fi/en/research/research-unit/

The opinions expressed in this paper are those of the authors and do not necessarily reflect the views of the Bank of Finland.

# Endogenous TFP, Business Cycle Persistence and the Productivity Slowdown

Michaela Schmöller\*

Martin Spitzer<sup>†</sup>

#### Abstract

This paper analyses the procyclicality of euro area total factor productivity and its role in business cycle amplification by estimating a medium-scale DSGE model with endogenous productivity mechanism on euro area data. Total factor productivity evolves endogenously as a consequence of costly investment in R&D and adoption of new technologies. We find that the endogeneity of TFP induces a high degree of persistence in the euro area business cycle via a feedback mechanism between overall economic conditions and investment in productivity-enhancing technologies. As to the sources of the euro area productivity slowdown, we conclude that a decrease in the efficiency of R&D investment is among the key factors generating the pre-crisis productivity slowdown, while starting from the Great Recession an increase in liquidity demand is identified as the most important driving force. The endogenous technology mechanism further exerts a dampening effect on the inflation response over the business cycle which helps rationalizing both the negligible fall in inflation during the Great Recession and the sluggish increase of inflation in the subsequent recovery.

JEL Classification: E24, E32, O31

**Keywords:** Endogenous Productivity, Euro Area Business Cycles, Weak Growth, Low Inflation

We would like to thank Diego Comin and Diego Rodriguez Palenzuela for valuable advice and support. We are also grateful to Gianluca Benigno, Jan Svejnar, Reinhilde Veugelers, Juha Kilponen, Tero Kuusi, Pertti Haaparanta, Antti Ripatti, Marcin Bielecki and Xiang Li who provided insightful discussions and comments. We further thank participants at the ECB seminars, European Network for Research and Investment Workshop at the European Investment Bank, Bank of Finland seminars, Working Group on Forecasting Meeting (Lisbon), FDPE Macroeconomics Workshop (Helsinki), 2018 Annual ESCB Research Cluster 2 Workshop at the Banque de France, Research Institute of the Finnish Economy seminar and the 19th IWH-CIREQ-GW Macroeconometric Workshop (Halle).

This paper should not be reported as representing the views of the European Central Bank or the Bank of Finland. The views expressed are those of the authors and do not necessarily reflect those of the European Central Bank or the Bank of Finland.

<sup>\*</sup> Bank of Finland; michaela.schmoller@bof.fi

<sup>&</sup>lt;sup>†</sup> European Central Bank; martin.spitzer@ecb.int

# Non-technical summary

The pre-crisis macroeconomic paradigm focused on short-term fluctuations around a given, predetermined trend in which technological progress is not explicitly modelled but instead evolves purely exogenously in the form of a technology shock. However, the experience of advanced economies starting from the financial crisis, characterized by a deep and highly persistent recession, slowing productivity growth and sluggish inflation, challenged existing modelling approaches. This experience highlighted, among others, the need for macroeconomic frameworks capable of making statements about the drivers of technological progress and productivity growth in a macroeconomic general equilibrium setting. This paper follows this approach by estimating a medium-scale DSGE model with an endogenous total factor productivity mechanism, as initially proposed by Anzoategui, Comin, Gertler and Martinez (2018) on euro area data to investigate the causes of the persistence of the Great Recession and the drivers of the productivity slowdown in the euro area.

We show that accounting for endogenous total factor productivity dynamics through productivityenhancing investments in R&D and technology adoption considerably increases the persistence of the euro area business cycle relative to standard macroeconomic frameworks in which TFP evolves purely exogenously. Moreover, we demonstrate that the endogenous part of total factor productivity is empirically of substantial magnitude and accounts for a large share of overall TFP in the euro area. Our findings point out that euro area TFP evolves clearly procyclically as it tends to increase in upswings and to fall in downturns.

As to the development of euro area TFP over time, we show that the productivity slowdown has commenced already before the Great Recession. Starting from the Great Recession, we document a pronounced further deceleration of productivity growth. Regarding the central factors driving the euro area productivity slowdown, we demonstrate that in the pre-crisis phase, a drop in R&D efficiency constitutes a main driving force. Since the Great Recession, the liquidity demand shock, which depresses consumption relative to safe asset holdings and the resulting fall in firms' technology adoption represents the most important influencing factor. In sharp contrast to standard macroeconomic models in which the supply side evolves strictly distinct from demand-side fluctuations, this result underlines the relevancy of demand shocks in explaining the evolution of the supply side in the euro area. We observe starting from 2015 a gradual improvement in the endogenous part of total factor productivity as a consequence of generally improving economic conditions. The observed reduced efficiency of R&D investment in producing new innovations, however, constitutes a drag on euro area TFP going forward.

Lastly, accounting for endogenous productivity dynamics features important inflation implications as the endogenous TFP mechanism dampens the inflation response over the business cycle and thus helps in explaining both the negligible decrease in euro area inflation during the crises vis-à-vis a marked output drop and the rather slow pickup during the post-crisis expansion.

## 1 Introduction

The recent recessions in the euro area were characterized by a severe and highly persistent output drop. Figure 1 demonstrates the marked fall in euro area real GDP starting from 2008. The persistence of the fall in output is striking as euro area GDP returned to pre-crisis levels only in 2015. In addition to the depth and severity of the recessions and the corresponding extent of business cycle persistence, the euro area experienced simultaneously a substantial slowdown in productivity. Figure 2 illustrates the evolution of hourly labor productivity in the euro area and shows that, while productivity growth had started to stagnate already in the pre-crisis phase, labor productivity fell drastically during the Great Recession. In the subsequent recovery, in turn, labor productivity has started to improve again, in line with general improving economic conditions. That given, the observed drop in labor productivity has been highly persistent and remains at this stage still below its pre-crisis trend.



Figure 1: Real GDP in the euro area; index (2010=100); source: Eurostat

This observed depth and persistence of the Great Recession and the prevalent slowdown in productivity in the euro area as well as other advanced economies have spurred theoretical studies, which explicitly model the evolution of productivity dynamics in macroeconomic models. More specifically, these frameworks take into account that, in contrast to the standard assumption in macroeconomic models, total factor productivity may not evolve purely exogenously driven only by standard technology shocks, but is instead subject to procyclical fluctuations and can thus constitute an important source of business cycle amplification. The procyclical nature of TFP has gained further attention in the context of the sluggish inflation response in the recent expansion, which has raised, among others, discussions on the appropriate assessment of the output gap following a deep recession (see Coibion, Gorodnichenko and Ulate (2017)), as well as potential driving forces counteracting inflationary pressures.



Figure 2: Hourly labor productivity in the euro area; index: 2010=100; source: Eurostat

Anzoategui, Comin, Gertler and Martinez (2018) provide a central contribution in the analysis of the procyclicality of total factor productivity and its role in business cycle amplification in the US economy. The underlying model framework constitutes the theoretical backbone of our model-based analysis of the depth of the recent recessions and the key drivers of the productivity slowdown observed in the euro area. The key insights by Anzoategui et al. (2018) can be summarized as follows. The study provides empirical evidence on the procyclical evolution of total factor productivity and its key driving forces research and development and technology adoption. Moreover, drawing on this evidence, they propose a medium-scale estimated DSGE model in the spirit of Smets and Wouters (2007), which features an endogenous total factor productivity mechanism. Hence, as opposed to standard macroeconomic models, total factor productivity is not limited to evolve purely exogenously in the form of a technology shock only, but instead evolves endogenously as the result of costly R&D investment and technology adoption. Anzoategui et al. (2018) show that taking into account the procyclicality of total factor productivity gives rise to a powerful mechanism of business cycle amplification capable of explaining the depth and persistence of the Great Recession in the US. Furthermore, the richness of the theoretical model permits the analysis of the central driving forces behind the US productivity slowdown, where the diminished efficiency of R&D investment is identified as the key source of slowdown in US productivity in the pre-crisis phase. In the period following the Great Recession, by contrast, the shock to liquidity demand, which favors safe asset holdings at the expense of consumption, emerges as the most important driver of the slowdown in US total factor productivity growth.

While the existing literature on the endogeneity of total factor productivity in macroeconomic models focuses on the US, evidence on the euro area is scarce. To shed light on the degree of procyclicality of euro area total factor productivity, its role in generating business cycle persistence and deep recessions, as well as the key drivers of the observed productivity slowdown, we estimate a medium-scale DSGE model with endogenous TFP mechanism as proposed by Anzoategui et al. (2018) on euro data. In doing so we provide insights in particular on the following central questions. Firstly, we investigate the nature of the euro area productivity slowdown as well as its key driving forces. We further provide empirical insights on the question to what degree total factor productivity in the euro area fluctuates procyclically over the business cycle. A crucial question addressed by our research centers around the feedback mechanism between the evolution of TFP on the one hand and overall economic conditions on the other hand and the role it accrues in explaining the severity and persistence of the recent euro area recessions. Lastly, based on our findings, we provide an outlook on the future evolution of euro area total factor productivity and its implications for the future dynamics of euro area GDP.

The central findings of this paper can be summarized as follows. Firstly, our results suggest that the endogenous modeling of total factor productivity substantially increases the persistence of the euro area business cycle relative to standard macroeconomic models with purely exogenous TFP. Crucially, we find that euro area total factor productivity exhibits a strong degree of procyclicality and that the corresponding resulting feedback mechanism between overall economic conditions on the one side and the evolution of total factor productivity on the other side constitutes a powerful channel capable of generating a high degree of business cycle persistence, providing important insights on both the depth and severity of the euro area recessions as well as on the drivers of the euro area productivity slowdown. More specifically, we find that the endogenous TFP component explains a high share of overall euro area total factor productivity. Regarding the evolution of the endogenous TFP component, our results show that it has exhibited a slowdown in growth discernible already in the early 2000s. During the Great Recession, however, this decline accelerated markedly and total factor productivity fell substantially. Concerning the main driving forces of the euro area productivity slowdown, we identify a decline in the efficiency of R&D investment in generating new technology innovations as the most important driver in the pre-crisis period. Starting from the Great Recession, in turn, we find that the liquidity demand shock, which raises savings at the expense of consumption and triggers typical business cycle dynamics in key economic variables, constitutes the most important endogenous source of the euro area productivity slowdown. Importantly, this result emphasizes the importance of demand-side shocks in explaining the evolution of euro area supply-side developments, while at the same time challenging the exogeneity of total factor productivity in standard macroeconomic frameworks in this context. As to the future outlook on the euro area productivity slowdown, we can deduct from our empirical analysis that towards the end of the sample, starting from approximately 2015 onward, that a recovery in the endogenous component of total factor productivity is discernible, fostered by the generally improving overall economic condition in the euro area, which support firms' activity in the field of R&D and technology adoption. A decline of the efficiency of R&D investment in triggering new innovations, as well as exogenous TFP shocks, however, remain drags on euro area productivity. Lastly, the analysis

of the euro area economy through the lens of a macroeconomic model in which total factor productivity evolves endogenously also holds important implications regarding the dynamics of euro area inflation in store as factoring in the spillover to productivity developments from overall economic conditions and vice versa mutes the response of inflation over the business cycle. In the event of a recessionary shock, the drop in TFP raises firms' production costs, which counteracts the standard inflationary tendencies and in sum dampens the fall in inflation. This property is thus in line with the empirically negligible downward adjustment of euro inflation during the Great Recession despite a major output drop. In an upswing, in turn, TFP increases in our euro area model as improving economic conditions support firms' investment in R&D and technology adoption, thus reducing inflationary pressures given reduced production costs as a result of the productivity gains. Thus, the endogenous TFP channel also contributes to explaining the sluggish response of inflation in the recent euro area upswing.

This paper is closely linked to studies which analyze the role of productivity dynamics in macroeconomic models and the corresponding feedback effects to and from overall economic conditions in this context. Bianchi, Kung and Morales (2019) represents a central study related to our analysis since their paper also investigates the relationship between cyclical fluctuations and productivity growth by means of an estimated DSGE model with endogenous productivity mechanism. In their framework productivity growth results from both knowledge accumulation in the form of vertical innovation (Aghion and Howitt (1992); Grossman and Helpman (1991)) through R&D and the degree of technology utilization. Their results also highlight the role of procyclical productivity movements in explaining business cycle persistence. Further analyses close to this paper studying the relationship between business cycle fluctuations and long-run growth are Comin, Gertler and Santacreu (2019) as well as Kung and Schmid (2015). Lastly, this paper relates to the literature on secular stagnation (Eggertsson, Mehrotra and Robbins (2019); Eggertsson, Mehrotra and Sing (2016); Gordon (2015)). Among these studies on secular stagnation, Benigno and Fornaro (2017) is particularly closely related as their analysis can be understood as a synthesis of the hysteresis view and the secular stagnation perspective as they demonstrate how at first temporary shortfalls in aggregate demand can translate into secular stagnation in a Keynesian growth model with nominal rigidities and technological innovation through research and development.

The outline of the paper can be summarized as follows. Firstly, we present the model framework we base our analysis on (section 2). Section 3 describes our estimation approach and shows the corresponding estimation results. We proceed to analyze the model dynamics and the key drivers of economic fluctuations in this framework (section 4). Section 5 presents the evolution of endogenous total factor productivity, the decomposition into its main components as well as its key underlying driving shocks. Section 6 outlines the implications for euro area macroeconomic policy. Section 7 concludes.

# 2 The model

Our analysis of the euro area business cycle and evolution of total factor productivity is based on the model proposed by Anzoategui, Comin, Gertler and Martinez (2018). The main model structure is a medium-scale New Keynesian DSGE model in the spirit of Christiano et al. (2005) and Smets and Wouters (2007) with the corresponding standard model features. The model entails price and wage rigidities of the Calvo type, monetary policy is modeled in the form of a Taylor rule and investment is subject to flow adjustment costs. As a central difference as opposed to standard DSGE frameworks in which TFP evolves purely exogenously, total factor productivity in this model is endogenous and the result of the creation of new technologies through R&D investment and technology adoption of these inventions in the production process. The model features two types of labor: Skilled labor employed in the R&D sector and in the technology adoption process, as well as unskilled labor used in production. While financial frictions are not explicitly modeled, the framework includes a liquidity demand shock, which exhibits transmission properties of a financial shock and induces typical business cycle comovement in key economic variables.

#### 2.1 Empirical evidence on the procyclicality of R&D and technology adoption

The theoretical model underlying our analysis is subject to procyclical productivity dynamics as a result of procyclical movements in R&D and technology adoption activity. This section shows empirical evidence on the procyclical evolution of both margins generating productivity advances in the theoretical model. We first address the procyclicality of R&D activities. Figure 3 presents the evolution of business R&D expenditure in the euro area and demonstrates that the observed pattern in R&D activity is clearly procyclical: Investment in R&D increases in upswings and declines in recessions. Importantly, we observe a pronounced fall in R&D investments during the Great Recession as well as the subsequent euro area debt crisis. Likewise, during the downturn in the early 2000s, R&D expenditures dropped markedly.

Let us now turn to the procyclicality of technological diffusion. Data on technology adoption are general scarce given the lack of series on the aggregate level. Anzoategui et al. (2018) use survey data on the speed of technology diffusion covering the fraction of companies which have adopted a certain technology for 26 production technologies over the time period 1947 to 2003 in the US and the UK. Based on these data, they estimate the cyclical effect on the speed of diffusion, controlling for variables governing the general process of technology diffusion and document a robust and positive effect of the cyclical position on the speed of technology adoption. Further, Anzoategui et al. (2018) demonstrate based on UK data for three internet-related technologies provided by Eurostat that the diffusion of these technologies substantially fell during the Great Recession and increased in the subsequent recovery. In addition to the pronounced procyclicality of technology diffusion, this study also emphasizes the high magnitude of fluctu-



Figure 3: R&D expenditures in the euro area (12 countries, fixed composition; log-linearly detrended data; source: Eurostat; data are deflated by the GDP deflator and population adjusted; see appendix A.1 for data sources). Shaded areas indicate OECD recession dates.

ations of technology adoption over the business cycle. Additionally, Anzoategui et al. (2018) approximate adoption activity by means of firms' expenditure on technology adoption. More specifically, they use a data set compiled by the Association of University Technology Managers on firms' expenditures to acquire licenses for the use of technologies generated by universities as well as research hospitals. They show that adoption expenditures are subject to a high degree of comovement with cyclical GDP. Moreover, adoption expenditures fell substantially during recessions and, most notably so, during the Great Recession. Having discussed the empirical evidence on the procyclicality of both research and development and adoption activities, we proceed to outline the main model structure underlying our analysis.

### 2.2 Production and endogenous TFP

This section presents the production structure of the model and demonstrates how total factor productivity and its endogenous component enter aggregate production in this framework. The model economy features two types of firms - intermediate goods and final goods producers. There are a continuum of measure one final goods producers, which are monopolistically competitive. Each final good firm *i* produces differentiated output  $Y_t^i$ . The corresponding final good composite  $Y_t$  is a CES aggregate of the differentiated final goods  $Y_t^i$ :

$$Y_t = \left(\int_0^1 \left(Y_t^i\right)^{\frac{1}{\mu_t}} di\right)^{\mu_t},$$
 (1)

where  $\mu_t > 1$  evolves exogenously. Final good firm *i* produces using  $X_t^i$  units of intermediate good composite, which constitutes the only production input, following the linear production

function:

$$Y_t^i = X_t^i. (2)$$

Final goods firms set its respective nominal price  $P_t^i$  on a staggered basis (see section (2.5.3)). Each intermediate goods firm produces a differentiated product and there exists a continuum of measure  $A_t$  of monopolistically competitive intermediate goods producers.  $A_t$  denotes the stock of adopted technologies, that is the stock of types of intermediate goods, which have been adopted for use in production. Each intermediate goods firm produces output  $X_t^j$ . The intermediate goods composite is a CES aggregate of the respective intermediate goods:

$$X_t = \left(\int_0^{A_t} \left(X_t^j\right)^{\frac{1}{\vartheta}} dj\right)^{\vartheta},\tag{3}$$

where  $\vartheta > 1$ .  $L_t^j$  denotes unskilled labor,  $K_t^j$  the stock of capital and  $U_t^j$  capital utilization, used by firm j respectively. Firm j produces output using unskilled labor  $L_t^j$  and utilization-adjusted capital services  $U_t^j K_t^j$  according to the Cobb-Douglas technology

$$X_t^j = \theta_t \left( U_t^j K_t^j \right)^\alpha \left( L_t^j \right)^{1-\alpha}.$$
(4)

We assume that intermediate goods firms can adjust their price in each period. Consequently, as opposed to final goods prices, intermediate goods prices are perfectly flexible. In a symmetric equilibrium for intermediate goods, we can derive the aggregate production function for the final good composite  $Y_t$  using equation (3) and (4) to a first order as:<sup>1</sup>

$$Y_t = \left[ (A_t)^{\vartheta - 1} \theta_t \right] (U_t K_t)^{\alpha} (L_t)^{1 - \alpha} .$$

$$\tag{5}$$

As equation (5) demonstrates, total factor productivity consists of both an exogenous component, i.e. the standard TFP shock,  $\theta_t$ , and an endogenous component  $(A_t)^{\vartheta-1}$ . Hence, the model is in addition to exogenous productivity variation also subject to endogenous productivity movements, generated by the expansion in the variety of adopted intermediate goods as measured by  $A_t$ . The model-inherent endogenous TFP mechanisms governing the creation of new goods and their adoption in the production process are described in the following section.

# 2.3 The endogenous TFP mechanism: Technological progress through R&D and technology adoption

The endogenous total factor productivity mechanism inherent to this model follows Comin and Gertler (2006). Hence, technological progress and thus productivity advances in this model are governed by a two-stage process, in the form of R&D on the one side and technology adoption on the other side. In the first stage, new technologies are invented through research and de-

<sup>&</sup>lt;sup>1</sup>See Anzoategui et al. (2018) for a detailed derivation of the aggregate production function.

velopment, adding to the total stock of technologies  $Z_t$ . In the subsequent stage of technology adoption, firms incorporate these technologies into production. Importantly, the creation of new technologies in itself does not improve productivity, as productivity improvements are only realized once firms adopt these technologies in their production processes. Hence, the model features a realistic distinction between innovation and technological adoption, which also empirically display different degrees of procyclicality. Concerning notation, we denote the stock of adopted technologies by  $A_t$ , which constitutes a subset of the total stock of technologies  $Z_t$ . Thus,  $Z_t - A_t$  denotes the stock of unadopted technologies. Innovators in the R&D sector create new technologies  $Z_t$ , while the adoption sector turns these technologies usable in production. This distinction between creation and adoption of technologies also enables to generate a realistic time lag between the invention of new goods and their adoption in production observed empirically. The stock of adopted technologies  $A_t$  corresponds to the endogenous component of TFP, as described in the previous section, and increases in  $A_t$  generate endogenous productivity growth in the model.

#### 2.3.1 R&D sector: Creation of new technologies

As previously outlined, innovation, i.e. the creation of new technologies, in the R&D sector constitutes the first stage generating endogenous productivity growth in this framework. The R&D sector features a continuum of measure one of innovators. Innovators employ skilled labor to create new intermediate goods.  $L_{srt}^p$  refers to skilled labor used by innovator p. Each unit of skilled labor employed in the R&D sector at time t can create  $\varphi_t$  new technologies in the subsequent period t + 1.  $\varphi_t$  evolves according to the process

$$\varphi_t = \chi_t Z_t L_{srt}^{p_z - 1},\tag{6}$$

where  $\chi_t$  denotes an exogenous shock to R&D technology and  $L_{srt}$  is the aggregate level of skilled labor working in research and development. More specifically,  $\chi_t$  denotes a R&D efficiency shock and hence captures exogenous variations in the efficiency of investment in R&D in creating new technologies. The presence of  $Z_t$ , which innovators take as given, incorporates public learningby-doing in the R&D process (Romer (1990)). The R&D process is subject to a congestion externality: Increases in aggregate R&D activity reduce R&D efficiency for each individual innovator as we assume  $\rho_z < 1.^2$ 

Innovator p's decision problem consists of choosing  $L_{srt}^p$  to maximize

$$\max_{L_{srt}^p} E_t \left\{ \Lambda_{t,t+1} J_{t+1} \varphi_t L_{srt}^p \right\} - w_{st} L_{srt}^p, \tag{7}$$

 $<sup>^{2}</sup>$ Anzoategui (2018) elaborate that the presence of a congestion externality enables constant returns to scale in the generation of new technologies. This property simplifies aggregation, while inducing diminishing returns for the aggregate.

where  $J_t$  denotes the value of an unadopted technology,  $\Lambda_{t,t+1}$  is the household's discount factor and  $w_{st}$  represents the real wage for a unit of skilled labor. Thus, the optimality condition for research and development can be stated as follows:

$$E_t \{\Lambda_{t,t+1} J_{t+1} \varphi_t\} - w_{st} = 0.$$
(8)

Using equation (6), the first order condition (8) implies

$$E_t \left\{ \Lambda_{t,t+1} J_{t+1} \chi_t Z_t L_{srt}^{\rho_z - 1} \right\} = w_{st}, \tag{9}$$

stating that the discounted marginal benefit from an additional unit of labor has to equal the marginal costs of skilled labor. Importantly, due to the procyclicality of profits in intermediate goods production, the value of an unadopted technology also evolves procyclically. This property implies in combination with the assumption of sticky wages a procyclical movement of skilled labor employed in the R&D sector  $L_{srt}$  and thus investment in research and development.<sup>3</sup>

We include the possibility that technologies become obsolete and denote the survival rate for a given technology by  $\phi$ . The stock of technologies  $Z_t$  evolves according to the following law of motion

$$Z_{t+1} = \varphi_t L_{srt} + \phi Z_t, \tag{10}$$

where  $\varphi_t L_{srt}$  measures the newly created technologies and  $\phi Z_t$  the surviving technologies in period t. Combining (6) and (10), we can derive growth of the technology stock as

$$\frac{Z_{t+1}}{Z_t} = \chi_t L_{srt}^{\rho_z} + \phi,$$
(11)

where  $\rho_z$  refers to the elasticity of the growth rate of technologies with respect to research and development activity, which is estimated (section 3).

#### 2.3.2 Adoption of new technologies

This section describes the adoption mechanism, i.e. the process of converting existing technologies  $Z_t$  to technologies usable in production  $A_t$ . Crucially, only if an invention is adopted by firms it will generate productivity advances. Moreover, incorporating an endogenous adoption process in the model aims at capturing the empirical properties that technology adoption takes on average time and that the adoption of new technologies varies procylically. Modeling technology adoption in the form of an adoption sector, permits the endogenous modeling of technological diffusion, while avoiding issues related to aggregation.<sup>4</sup> Thus, a competitive group of adopters converts unadopted technologies into technologies usable in production. More

 $<sup>{}^{3}</sup>$ R&D investment corresponds to the wage bill of skilled labor allocated to this sector  $w_{st}L_{srt}$ .

<sup>&</sup>lt;sup>4</sup>More concretely, this approach avoids the need of keeping track of the fraction of firms which have adopted each of the respective technologies.

specifically, adopters buy the rights to use the technology at the competitive price  $J_t$ , which corresponds to the value of an unadopted technology. The conversion of technologies into technologies which can be used in production requires input of skilled labor and takes time on average, where the conversion rate is subject to endogenous variation. The probability  $\lambda_t$  that an adopter successfully renders a product usable in any period equals to

$$\lambda_t = \lambda \left( Z_t L_{sat} \right),\tag{12}$$

where  $L_{sat}$  refers to skilled labor employed in technology adoption.<sup>5</sup> We assume  $\lambda' > 0$  and  $\lambda'' < 0$ , which implies that the pace of adoption is an increasing and concave function of skilled labor allocated to this sector. Note that the specification of the adoption process entails that technological diffusion does not occur instantaneously but takes time on average. Let  $\bar{\lambda}$  denote the steady state value of  $\lambda_t$ , then the average time until a new technology is adopted corresponds to  $\frac{1}{\lambda}$ . Outside the steady state, the speed of adoption varies with the input of skilled labor  $L_{sat}$ . Note also that the adoption process is subject to a spillover effect from the total technology stock  $Z_t$ , implying that the efficiency of the adoption process increases in line with the state of the technology in the economy. This assumption guarantees the existence of a balanced growth path as the efficiency of the adoption process has to increase when the stock of technologies requiring adoption expands.

Once a technology has been rendered usable in production, the adopter sells the rights to use the technology to a monopolistically competitive intermediate goods producer which uses the good according to equation (4). Let  $\Pi_t^x$  denote the profit of an intermediate good firm from producing the good. The adopter sells the adopted technology at the competitive price  $V_t$ , which equals to the present discounted value of profits from producing the good

$$V_t = \Pi_t^x + \phi E_t \left\{ \Lambda_{t,t+1} V_{t+1} \right\}.$$
(13)

The adopter's maximization problem consists of choosing  $L_{sat}$  to maximize the value of an unadopted technology  $J_t$ :

$$J_{t} = \max_{L_{sat}} E_{t} \left\{ -w_{st} L_{sat} + \phi \Lambda_{t,t+1} \left[ \lambda_{t} V_{t+1} + (1 - \lambda_{t}) J_{t+1} \right] \right\}$$
(14)

subject to condition (12).  $w_{st}L_{sat}$  refers to total adoption expenditures, i.e. the expenditures on skilled labor allocated in the adoption process, while the second term in equation (14) denotes the discounted gains, which equals to the sum of adopted and unadopted technologies weighted by their respective probability. The corresponding first order condition can be expressed as follows:

$$Z_t \lambda'_t \phi E_t \{ \Lambda_{t,t+1} [V_{t+1} - J_{t+1}] \} = w_{st}.$$
(15)

<sup>&</sup>lt;sup>5</sup>In estimating the model, we employ the functional form of the adoption rate as follows:

 $<sup>\</sup>lambda(\bullet) = \kappa_{\lambda}(\bullet)^{\rho_{\lambda}}$ , where  $\kappa$  and  $\rho_{\lambda}$  are parameters and  $0 < \rho_{\lambda} < 1$  applies.

Hence, marginal costs of adoption  $w_{st}$  equal the marginal gain from adoption expenditures, i.e. the increase in the adoption probability times the discounted difference between the value of an adopted versus unadopted technology. Let us now turn to the evolution of adoption activity over the business cycle. Firstly, note that the term  $V_t - J_t$  varies procyclically as future profits increase the value of adopted technologies  $V_t$  more strongly than the value of unadopted technologies  $J_t$ . From this property combined with stickiness in  $w_{st}$  follows the procyclicality of  $L_{sat}$ . Consequently, also the speed of adoption  $\lambda_t$  will evolve procyclically.

Since  $\lambda_t$  is not subject to adopter-specific characteristics, it is possible to sum across adopters in order to derive the law of motion of adopted technologies:

$$A_{t+1} = \lambda_t \phi \left[ Z_t - A_t \right] + \phi A_t, \tag{16}$$

where  $Z_t - A_t$  corresponds to the stock of unadopted technologies. Thus, tomorrow's stock of technology incorporated in production - and thus the endogenous component of TFP - equals to the sum of the stock of non-obsolete technologies surviving in the current period  $\phi A_t$  and the technologies newly adopted in this period  $\lambda_t \phi [Z_t - A_t]$ .

#### 2.4 Households

The representative household saves in capital and riskless bond which are in zero net supply. The household rents capital to intermediate goods firms. Further, the model is subject to habit formation in consumption and households act as monopolistically competitive suppliers of differentiated types of labor. In addition to these standard properties, the household's problem is subject to the following, additional features. Firstly, the household supplies two types of labor: Skilled labor  $L_{st}^h$  used both in research and development and adoption and unskilled labor  $L_{ut}^h$  employed in the production of intermediate goods. Moreover, we assume that the household has a preference for holding the safe asset, which can be interpreted as a preference for liquidity. More specifically, bond holdings enter directly the utility function.<sup>6</sup> Further, we assume a liquidity demand shock  $\varrho_t > 0$  which features transmission properties of a financial shock.<sup>7</sup>  $C_t$  denotes consumption,  $B_t$  riskless bond holdings and  $\Pi_t$  profits from ownership of monopolistically competitive firms.  $K_t$  refers to capital,  $Q_t$  to the price of capital,  $R_t^k$  to the rate of return on capital and  $D_t$  to the rental rate of capital. The households' decision problem can be stated as follows:

$$\max_{C_{t},B_{t+1},K_{t+1},L_{ut}^{h},L_{st}^{h}} E_{t} \sum_{\tau=0}^{\infty} \beta^{\tau} \left\{ log \left( C_{t+\tau} - bC_{t+\tau-1} \right) + \varrho_{t} B_{t+1} - \left[ \frac{v_{u} \left( L_{t}^{h} \right)^{1+\varphi} + v_{s} \left( L_{st}^{h} \right)^{1+\varphi}}{1+\varphi} \right] \right\}$$
(17)

<sup>&</sup>lt;sup>6</sup>This step follows Krishnamurthy and Vissing-Jorgensen (2012).

<sup>&</sup>lt;sup>7</sup>The assumption of a shock to liquidity demand follows Fisher (2015). Accordingly, the liquidity demand shock corresponds to an explicit formulation of the risk shock in Smets and Wouters (2007).

subject to

$$C_t = w_{ut}^h L_{ut}^h + w_{st}^h L_{st}^h + \Pi_t + R_t^k Q_{t-1} K_t + R_t B_t - B_{t+1},$$
(18)

where  $R_t^k = \frac{D_t Q_t}{Q_{t-1}}$ . Note that  $\Lambda_{t,t+1}$  denotes the stochastic discount factor of the household given by  $\Lambda_{t,t+1} \equiv \frac{\beta u'(C_{t+1})}{u'(C_t)}$ , where  $u'(C_t) = \frac{1}{C_t - bC_{t-1}} - \frac{b}{C_{t+1} - bC_t}$ . Lastly, let  $\zeta_t$  be the liquidity preference shock denoted in units of the consumption good  $\zeta_t \equiv \frac{\rho_t}{u'(C_t)}$ . Thus, the first order conditions for capital and the riskless bond can be respectively derived as

$$1 = E_t \{ \Lambda_{t,t+1} R_{kt+1} \}$$
(19)

$$1 = E_t \{ \Lambda_{t,t+1} R_{t+1} \} + \zeta_t.$$
(20)

Equation (20) shows that the presence of the liquidity demand shock induces a distortion to the first order condition for the riskless bond. An increase in  $\zeta_t$  affects the economy similar to an increase in risk: For a given riskless rate  $R_{t+1}$ , a rise in  $\zeta_t$  triggers a precautionary savings effect in the sense that households lower their consumption in order to obey the first order condition via a fall in  $\Lambda_{t,t+1}$ . Furthermore, the shock to liquidity demand lowers investment demand since, as described by equation (19), the drop in  $\Lambda_{t,t+1}$  causes an increase in the return on capital. The decrease of the stochastic discount factor also induces a decrease of R&D and adoption, as implied by equation (9) and (15). In conclusion, the liquidity demand shock  $\zeta_t$  induces a positive comovement between consumption and investment. This can be further demonstrated by combining equation (19) and (20)

$$E_t \{ \Lambda_{t,t+1} \left( R_{kt+1} - R_{t+1} \right) \} = \zeta_t, \tag{21}$$

which states that a rise of  $\zeta_t$  increases the interest rate spread  $R_{kt+1} - R_{t+1}$  and thus highlights the similarity in terms of transmission mechanism to a financial shock.<sup>8</sup>

#### 2.5 Standard DSGE model features

This section presents further key features of the model. For tractability, we focus on the resulting equilibrium conditions.

#### 2.5.1 Intermediate goods firms: factor demands

This section outlines the factor demands resulting from the first order conditions of intermediate goods firms. Intermediate goods producers choose capital  $K_t^j$ , utilization  $U_t^j$  and labor  $L_t^j$  to minimize costs given the relative price of intermediate goods composite  $p_t^x$ , the price of capital  $Q_t$ , the rental rate  $D_t$ , the real wage  $w_t$  and the desired markup  $\varsigma$ . The capital utilization decision

<sup>&</sup>lt;sup>8</sup>Anzoategui et al. (2018) provide empirical evidence for the high correlation between the model-implied liquidity demand shock and credit spreads.

is endogenized by assuming that the capital depreciation rate  $\delta\left(U_t^j\right)$  is increasing and convex in capital utilization  $U_t^j$ .<sup>9</sup> The first order conditions resulting from the firm's cost minimization problem for  $K_t^j$ ,  $U_t^j$  and  $L_t^j$  respectively can be stated as:

$$\alpha \frac{p_t^x X_t^j}{K_t^j} = \varsigma \left[ D_t + \delta \left( U_t^j \right) Q_t \right]$$
(22)

$$\alpha \frac{p_t^x X_t^j}{U_t^j} = \varsigma \delta' \left( U_t^j \right) Q_t K_t^j \tag{23}$$

$$(1-\alpha)\frac{p_t^x X_t^j}{L_t^j} = \varsigma w_{ut}.$$
(24)

The desired markup  $\varsigma$  is assumed to be lower than the optimal unconstrained markup  $\vartheta$  due to the potential threat of entry of imitators as commonly assumed by the literature.

#### 2.5.2 Capital producers: investment

New capital goods are created using final output by competitive capital producers. The capital producers sell the newly created capital goods to households, which, in turn, rent the capital to firms. We denote by  $I_t$  the newly produced capital, by  $p_t^x$  the replacement price of capital, i.e. the relative price of converting final output into new capital, and by  $\gamma_y$  the growth rate of  $I_t$  in the steady state. Further, investment is subject to flow adjustment costs.<sup>10</sup> We assume an increasing and concave adjustment cost function  $f\left(\frac{I_t}{(1+\gamma_y)I_{t-1}}\right)$  with the properties f(1) = f'(1) = 0 and f''(1) > 0. The corresponding first order gives Tobin's Q of investment, which states the ratio of the market value to capital to the replacement price:

$$\frac{Q_t}{p_t^k} = 1 + f\left(\frac{I_t}{(1+\gamma_y)I_{t-1}}\right) + \frac{I_t}{(1+\gamma_y)I_{t-1}}f'\left(\frac{I_t}{(1+\gamma_y)I_{t-1}}\right) \\
-E_t\Lambda_{t,t+1}\left(\frac{I_{t+1}}{(1+\gamma_y)I_t}\right)^2 f'\left(\frac{I_{t+1}}{(1+\gamma_y)I_t}\right),$$
(25)

where  $log(p_t^k)$  follows an AR(1) process subject to the parameters  $\rho_{pk}$  and  $\sigma_{pk}$ . The following equation presents the law of motion for capital

$$K_{t+1} = I_t + (1 - \delta(U_t)) K_t.$$
(26)

#### 2.5.3 Price and wage setting

As in Smets and Wouters (2007), both nominal prices and wages are subject to staggered price setting according to Calvo adjustment rules. We denote the probability a firm cannot adjust its

<sup>&</sup>lt;sup>9</sup>In endogenizing capital utilization, Anzoategui et al. (2018) follow the approach of Greenwood et al (1988). <sup>19</sup>Th is approach follows: Christians et al. (2005)

<sup>&</sup>lt;sup>10</sup>This approach follows Christiano et al. (2005).

price by  $\xi_p$  and the probability a firm cannot adjust its wage by  $\xi_w$ .  $\iota_p$  refers to the degree of indexation of prices to past inflation, while  $\iota_w$  is the corresponding counterpart for wages. As previously outlined, households supply two types of labor (unskilled and skilled). We assume both types of labor to be subject to the identical frequency of wage adjustment. Let us denote by  $\pi_t$  the inflation rate and by  $mc_t$  the marginal cost of final good producers, expressed in log deviation from steady state. From here, the price Phillips curve can be derived as

$$\pi_t = \kappa m c_t + \frac{\iota_p}{1 + \iota_p \tilde{\beta}} \pi_{t-1} + \frac{\beta}{1 + \iota_p \tilde{\beta}} E_t \left[ \pi_{t+1} \right] + \epsilon \mu_t, \tag{27}$$

where  $\tilde{\beta} = \frac{\beta}{1+\gamma_y}$  and  $\kappa = \frac{(1-\xi_p \tilde{\beta})(1-\xi_p)}{\xi_p(1+\iota_p \tilde{\beta})}$ .  $\varepsilon \mu_t$  denotes a shock to final goods markup, which follows an AR(1) process with parameters  $\rho_{\mu}$  and  $\sigma_{\mu}$ . The Phillips curve for unskilled wages is given by

$$(1+\kappa_w)\tilde{w_t^u} = \frac{1}{1+\tilde{\beta}} \left( w_{t-1}^{\tilde{u}} + \iota_w \pi_{t-1} - \left(1+\tilde{\beta}\iota_w\right)\pi_t \right) + \frac{\tilde{\beta}}{1+\tilde{\beta}} E_t \left[ \tilde{w_{t+1}} + \pi_{t+1} \right] \\ + \kappa_w \left( \tilde{muc}_t - \varphi \tilde{l}_t^u \right) + \epsilon \mu w_t,$$

$$(28)$$

where  $\kappa_w = \frac{(1-\xi_w)(1-\xi_w\tilde{\beta})}{\xi_w(1+\tilde{\beta})(1+\varphi(1+\frac{1}{mu_w}))}$  and  $\bar{\mu}_w$  refers to to the steady state wage mark-up. The variables  $\tilde{muc}$ ,  $\tilde{w}$ , l denote the marginal utility of consumption, as well as the unskilled wage and hours in log deviation from steady state respectively.  $\epsilon \mu w_t$  is a shock to the wage mark-up, which follows an AR(1) process with parameters  $\rho_{\mu w}$  and  $\sigma_{\mu w}$ . Replacing unskilled wages and labor for their skilled equivalents, the wage Phillips curve for skilled wages is identical.<sup>11</sup>

#### 2.5.4 Monetary policy

Monetary policy is modelled in the form of a nonlinear Taylor rule which includes a zero lower bound constraint. More specifically, the central bank conducts monetary policy by setting the nominal interest rate  $R_{nt+1}$  according to

$$R_{t+1}^{n} = r_t^m \left( \left(\frac{\pi_t}{\pi^0}\right)^{\phi_\pi} \left(\frac{L_t}{L^{ss}}\right)^{\phi_y} R^n \right)^{1-\rho^R} \left(R_t^n\right)^{\rho^R}$$
(29)

subject to the zero lower bound constraint

$$R_{t+1}^n \ge 1,\tag{30}$$

where  $R^n$  denotes the steady state nominal rate,  $\pi^0$  the central bank's inflation target,  $L_t$  total employment and  $L^{ss}$  employment in the steady state. Further,  $\phi_{\pi}$  denotes the weight attributed to inflation gap, while  $\phi_y$  refers to the weight on the capacity utilization gap in the Taylor rule.

<sup>&</sup>lt;sup>11</sup>Note that in the model estimation, only unskilled wage setting is subject to wage markup shocks while the markup for skilled labor is assumed to be constant at its steady state level.

Lastly,  $r_t^m$  follows an AR(1) process subject to the parameters  $\rho^{mp}$  and  $\sigma^{mp}$ .

#### 2.5.5 Aggregation and equilibrium

The aggregate resource constraint can be derived as

$$Y_{t} = C_{t} + p_{kt} \left[ 1 + f \left( \frac{I_{t}}{(1 + \gamma_{y}) I_{t-1}} \right) \right] I_{t} + G_{t},$$
(31)

where  $G_t$  denotes government consumption.  $G_t$  is financed by lump sum taxes and follows the AR(1) process

$$\log\left(\frac{G_t}{\left(1+\gamma_y\right)^t}\right) = \left(1-\rho_g\right)\bar{g} + \rho_g \log\left(\frac{G_{t-1}}{\left(1+\gamma_y\right)^{t-1}}\right) + \epsilon_t^g.$$
(32)

Lastly, the labor market for skilled labor must clear in equilibrium:

$$L_{st}^h = L_{sat} + L_{srt}. (33)$$

## 3 Estimation

We estimate this DSGE model with endogenous TFP mechanism as proposed by Anzoategui et al. (2018) on euro area data using Bayesian methods. Estimation is performed on quarterly data from 1999:I to 2007:IV. Due to the exceptional circumstances of the Great Recession and potential concomitant biases resulting from the crisis episode, we exclude the time period from 2008:I onward in the model estimation. The broader model-based analysis, by contrast, is based on the full sample (1999:I to 2017:II). We estimate the model using eight time series: The growth rates of respectively GDP, consumption, investment, wages, hours worked and R&D expenditures, as well as the series on inflation and the nominal risk-free rate. We define euro area in terms of the fixed composition concept (EA-12). The mixed-frequency of the data arising from annual data on R&D expenditure is accounted for by taking a Kalman filter approach. Appendix A.1 presents a detailed description of the data used in estimation.

We estimate the following technology parameters in the model. Regarding the key technological parameters in our model, we estimate the elasticity of the creation of new technologies with respect to research and development  $\rho_z$ . Further, by estimating the steady state growth rate of the economy  $g_y$ , also the average productivity of R&D  $\bar{\chi}$  is pinned down. Moreover, following Anzoategui et al. (2018), all standard DSGE parameters except for the final goods markup are estimated.<sup>12</sup>

<sup>&</sup>lt;sup>12</sup>The underlying intuition for calibrating the final goods markup are potential identification issues related to an additional markup in intermediate goods production.

#### 3.1 Calibrated parameters

Table 1 presents an overview of the calibrated parameters and their respective values. In calibrating the model we closely follow the calibration approach by Anzoategui et al. (2018) for the US economy. We set the markups on intermediate ( $\varsigma$ ) and final goods ( $\mu$ ) to 1.18 and 1.1 respectively.<sup>13</sup> We calibrate the elasticity of substitution in intermediate goods  $\vartheta$  to 1.37.<sup>14</sup> We calibrate the steady state government expenditure to GDP ratio  $\frac{G}{Y}$  to 0.2 and the steady state depreciation rate  $\delta$  to 0.02.

Regarding the parameters governing the endogenous technological process we calibrate the steady state adoption lag  $\bar{\lambda}$ , the obsolescence rate  $(1 - \phi)$  as well as the elasticity of the adoption probability  $\lambda$  with respect to adoption expenditures  $\rho_{\lambda}$ . More specifically, we set  $\bar{\lambda}$  to generate an average adoption lag equal to 5 years.<sup>15</sup> We further calibrate an obsolescence rate  $1 - \phi$  of 2% (quarterly).<sup>16</sup> Lastly, we set the elasticity of adoption with respect to skilled labor  $\rho_{\lambda}$  to  $0.95.^{17}$ 

Parameter	Description	Value
δ	Capital depreciation	0.0200
$\frac{G}{Y}$	Steady state government consumption/output ratio	0.2000
$\mu$	Steady state final goods mark up	1.1000
ς	Steady state intermediate goods mark up	1.1800
θ	Intermediate goods elasticity of substitution	1.3699
$1-\phi$	Obsolescence rate	0.0200
$\overline{\lambda}$	Steady state adoption lag	0.0500
$\rho_{\lambda}$	Adoption elasticity	0.9500

Table 1: Calibrated parameter values

<sup>&</sup>lt;sup>13</sup>The ratio of R&D to GDP is increasing in the size of the markups and decreasing in  $\rho_{\lambda}$ . Thus, Anzoategui et al. (2018) propose a calibration choice of the markups at the lower end of the literature (see Jaimovich (2007)) in order to ensure a conservative calibration of  $\rho_{\lambda}$ .

 $<sup>^{14}</sup>$ This value is very similar to the parameter choice of 1.35 in Anzoategui et al. (2018) which was calibrated to generate an elasticity of substitution of 3.85 between intermediate goods which is consistent with the estimates by Broda and Weinstein (2006).

<sup>&</sup>lt;sup>15</sup>Anzoategui et al. (2018) propose this parameter choice as it is in line with the estimates in Cox and Alm (1996), Comin and Hobijn (2010) and Comin and Mestieri (2015).

 $<sup>^{16}</sup>$ Anzoategui et al. (2018) point out that the set value corresponds to the average of the estimated obsolescence rate as estimated in terms of patent renewal rates (Bosworth (1978)) and the rate of decay of patent citations (Caballero and Jaffe (1993)).

<sup>&</sup>lt;sup>17</sup>This calibration of  $\rho_{\lambda}$  is close to but slightly higher than in Anzoategui et al. (2018) in order to take into account the lower R&D to GDP ratio in the euro area as opposed to the US case.

#### 3.2 Estimation results

Table 2 shows the prior and posterior distributions from the Bayesian estimation of our model parameters. For the standard parameters, we follow closely the choice of priors by Anzoategui et al. (2018) and Smets and Wouters (2007). Further, we estimate the elasticity of the R&D parameter  $\rho_z$  based on the prior proposed by Anzoategui et al. (2018) which follows a beta distribution centered around mean 0.6 (Griliches (1990)). The estimated standard model parameters are in line with the findings of the overall literature. Importantly, our estimate of the R&D elasticity  $\rho_z$  is in line with the Griliches (1990) estimates.

Parameter	Description	Dist	Prior		Pos	Posterior	
			Mean	St.Dev.	Mean	St.Dev.	
$\rho^R$	Taylor rule smoothing	Beta	0.70	0.15	0.885	0.0384	
$\phi_{\pi}$	Taylor rule inflation	$\operatorname{Gamma}$	1.50	0.25	1.320	0.2294	
$\phi_y$	Taylor rule labour	$\operatorname{Gamma}$	0.30	0.10	0.348	0.1110	
$\varphi$	Inverse Frisch elast.	$\operatorname{Gamma}$	2.00	0.75	2.615	0.8314	
f''	Investment adj. cost	Gamma	4.00	1.00	3.741	0.8025	
$\frac{\delta'(U)}{\delta}$	Capital util. elas.	Gamma	4.00	1.00	4.148	0.9983	
$\xi_p$	Calvo prices	Beta	0.50	0.10	0.562	0.1111	
$\xi_w$	Calvo wages	Beta	0.50	0.10	0.654	0.0960	
$\iota_p$	Price indexation	Beta	0.50	0.15	0.204	0.0970	
$\iota_w$	Wage indexation	Beta	0.50	0.15	0.216	0.0937	
$\mu_w$	SS Wage markup	Normal	0.15	0.05	0.085	0.0591	
b	Consumption habit	Beta	0.70	0.10	0.599	0.0523	
$\rho_z$	R&D elasticity	Beta	0.60	0.15	0.540	0.1233	
$\beta^{est}$	$100 \times (\beta^{-1} - 1)$	$\operatorname{Gamma}$	0.25	0.10	0.392	0.1145	
$\alpha$	Capital share	Normal	0.30	0.05	0.248	0.0443	
$100 * \gamma_y$	SS output growth	Normal	0.10	0.20	0.301	0.0359	

Table 2: Prior and posterior distributions of estimated parameters

## 4 Model dynamics and key drivers of economic fluctuations

We proceed to give an overview of the main model mechanisms and key drivers of economic fluctuations in this framework. Section (4.1) demonstrates which shocks are key in driving economic fluctuations. Section (4.2) presents the model dynamics following a shock to liquidity demand in the presence of the endogenous TFP mechanism as opposed to the model with purely exogenous total factor productivity.

Variables	Liquidity	Money	$\operatorname{Govt}$	Price of	$\mathrm{TFP}$	R&D	Mark up	Wage
	Demand		$\operatorname{Exp}$	Capital				mark up
Output Growth	64.46	12.69	16.04	3.08	1.49	0.01	1.85	0.39
Consumption Growth	77.57	13.89	4.89	0.11	1.51	0.00	1.62	0.40
Investment Growth	39.84	11.29	3.22	40.44	1.57	0.03	2.90	0.72
Inflation	0.84	0.91	0.22	0.06	7.99	0.06	61.14	28.78
Nominal R	33.44	39.90	1.60	0.84	3.34	0.04	12.77	8.08
Hours	63.88	16.02	7.61	2.48	6.91	0.03	1.61	1.45
Endogenous TFP	71.62	15.84	0.70	0.46	0.79	0.73	9.72	0.13

Table 3: Variance decomposition

#### 4.1 Sources of model variation

Table 3 shows the variance decomposition of the key model variables.<sup>18</sup> The central finding is that the liquidity demand shock, interpretable as a shock which depresses consumption and favors safe asset holdings, represents the most important source of variation in this model: It accounts for more than 60% of the variation in output growth, for roughly 75% and 40% of the variation in consumption and investment growth respectively and for over 60% of the variation in hours worked. Crucially, demand shocks play an important role in explaining the variation in the endogenous total factor productivity component implying that demand side shocks exert important effects on the supply side. More specifically, 71.6% of the variation in endogenous total factor productivity are accounted for by the liquidity demand shock and 15.8% by the monetary policy shock. By contrast, the standard TFP shock, i.e. the shock to the exogenous component of total factor productivity, plays only a negligible role in explaining the variation of the key economic variables.

We now turn to the analysis of the historically key drivers of recessions. Figure 4 illustrates the historical development of GDP growth and the contributions of the standard TFP shock and the liquidity shock respectively. The historical decomposition demonstrates that the liquidity demand shock constitutes clearly the central driver of the recent recessions in the euro area. Technology shocks, instead, display a subordinate role in explaining recessions. This further supports the role of the liquidity demand shock in explaining economic fluctuations and the most important driver of recessions in the euro area.

<sup>&</sup>lt;sup>18</sup>We take a shadow rate approach as the shadow rate provides a more comprehensive measure of the monetary policy stance, including the role of non-standard monetary policy measures. As a robustness check, we use an alternative non-linear model specification which imposes the ZLB constraint from which we extract smoothed variables and corresponding shock contributions using the iterative method developed by Anzoategui (2017). The corresponding results are highly similar to the findings in our baseline model, where the key difference constitutes the somewhat higher role attributed to the liquidity demand shock in explaining economic fluctuations in the nonlinear model, resulting from the inactivity of monetary policy at the zero lower bound.



Figure 4: Smoothed shocks from model with endogenous TFP mechanism. Data used as described in Appendix A.1.

# 4.2 Impulse response analysis: Endogenous TFP and business cycle persistence

This section illustrates the model dynamics and demonstrates the impulse responses following a shock to liquidity demand. We focus on the latter due to its high relevancy in explaining economic fluctuations and as it triggers a typical business cycle comovement of key economic variables (see section 4.1). Figure 5 shows the impulse responses of the key model variables to a one standard deviation liquidity demand shock. A rise in the demand for safe asset holdings brought about by the liquidity demand shock generates a fall in consumption and reduces the holding of the risky asset. Consequently, this depresses the safe real rate  $R_{t+1}$  and exerts upward pressure on the return to capital  $R_{t+1}^k$ . Capital investment falls. Importantly, the drop in aggregate demand following the liquidity demand shock depresses firm profits and consequently productivity enhancing investments, i.e. investment in R&D and technology adoption. Crucially, the model with endogenous total factor productivity dynamics displays a markedly stronger degree of business cycle persistence vis-à-vis a standard DSGE model with exogenous productivity (red dashed line). More specifically, the output drop triggered by a liquidity demand shock is substantially stronger and also more persistent under endogenous TFP dynamics. The underlying cause is the presence of a strong feedback mechanism between the evolution of TFP on the one side and overall economic conditions on the other side, which is absent in standard macroe-



Figure 5: Impulse response to a 1 standard deviation liquidity demand shock

conomic frameworks: The corresponding drop of both R&D and technology adoption activity following an adverse liquidity demand shock depresses total factor productivity, reinforcing the initial output drop. As a result, the presence of the endogenous TFP mechanism is capable of generating deep and persistent recessions and our estimated DSGE model with endogenous TFP mechanism can contribute both to explaining the depth and persistence of the recent euro area recessions and to the analysis of the sources underlying the euro area productivity slowdown.

# 4.3 Inflation implications: Muted inflation response due to the interaction of inflation and productivity dynamics

Let us now turn to the implications for inflation dynamics resulting from the presence of the endogenous productivity mechanism and the related interaction between inflation and productivity dynamics. Figure 6 illustrates the evolution of euro area core inflation and real GDP over the business cycle. It is striking that during the Great Recession, the downward adjustment of inflation was of minor magnitude vis-à -vis the marked drop in output. Moreover, also in the subsequent expansion, the observed reaction in inflation has been subdued in light of the sustained period of sound economic growth. Our model can match the observed patterns of inflation observed over the past decade: The response of inflation over the business cycle is substantially muted as opposed to the standard model with exogenous total factor productivity as a result of the model-inherent interaction of inflation and productivity dynamics. More



Figure 6: Inflation and GDP growth in the euro area (source: Eurostat; core inflation: HICP, annual rate of change; overall index excluding energy and unprocessed food; GDP growth: calendar and seasonally adjusted, chained volumes)

specifically, the reaction of inflation to a contractionary demand shock is negative, as is the case in standard New Keynesian DSGE models. However, a downturn will also be accompanied by a deceleration in productivity, raising price pressures and hence dampening the downward adjustment in inflation. Likewise, in the case of an expansionary liquidity demand shock, inflation would increase by less than in the framework with exogenous productivity as the price pressure resulting from increased demand are partly offset by decreased price pressures due to productivity improvements. Hence, the endogeneity of TFP in the model can also provide useful insights on potential drivers of the puzzling evolution of euro area inflation during the Great Recession and the subsequent recovery: The dampening effect resulting from the endogeneity of total factor productivity may have contributed to both the modest downward adjustment of inflation during the Great Recession and the subdued inflation response in the recent euro area expansion.

#### 4.4 The role of the ZLB

Let us now address the implications of the presence of the zero lower bound constraint on the central bank's policy rate on economic dynamics, business cycle amplification and productivity developments. Figure 7 shows the economy's reaction to a large recessionary liquidity demand shock which induces the zero lower bound to bind.<sup>19</sup> Accordingly, the central effects of a binding zero lower bound can be summarized as follows. As the shock in the economy is large and thus the corresponding shortfall in demand sufficiently pronounced, the required nominal interest rate in

<sup>&</sup>lt;sup>19</sup>The shock illustrated in Figure 7 constitutes a large shock in the sense that it corresponds to a 15 standard deviations liquidity demand shock.



Figure 7: Effects of the ZLB in business cycle amplification

line with the central bank's policy rule falls below zero. With the optimal monetary policy rate in negative territory, a binding zero lower bound will constitute a significant obstacle to monetary policy in economic stabilization. The right uppermost column in Figure 7 demonstrates that the fall in total factor productivity will be even more marked relative to the linear endogenous DSGE model due to the binding constraints on the central bank's capability in policy intervention. This translates into a higher drop of aggregate demand, which depresses firm profits and thus the value of an unadopted and adopted technology respectively. The latter reduce research and development as well as firms' technology adoption activity, inducing a more pronounced fall in total factor productivity than in the linear baseline model. Crucially, a binding zero lower bound translates into a substantially larger drop in economic output. The first and obvious channel is the more severe shortfall in demand when monetary policy is constrained and hence ineffective in economic stabilization. The second underlying driver is the more intense decrease of total factor productivity as a reaction to a recessionary shock. In conclusion, both the scale and the persistence of the output drop are substantially larger in the case when monetary policy is constrained, emphasizing the role accruing to monetary policy in economic stabilization in this context. Crucially, in the presence of the endogenous TFP mechanism presented in section 4.2, a binding zero lower bound will exert more severe consequences than implied by standard macroeconomic workhorse models, due to the increased degree of business cycle persistence and resulting hysteresis effects in productivity. Thus, taking into account the endogeneity of productivity developments increases the negative impact of the ZLB due to its adverse impact

not only on the demand side but also on the evolution of the supply side.

# 5 The euro area productivity slowdown: Evolution and key drivers of total factor productivity

This section analyses the productivity slowdown in the euro area from the perspective of this model. We focus on the evolution of the endogenous component of TFP, its empirical magnitude, as well as the main determinants of euro area total factor productivity. Based on these findings we seek to give insights on the key drivers of euro area total factor productivity and hence the productivity slowdown. Let us first address how model-implied productivity relates to empirically observable data. Recall from equation (5) that total factor productivity equals to  $(A_t)^{\vartheta-1}\theta_t$  and thus consists of an exogenous component  $\theta_t$ , i.e. a standard TFP shock, and an endogenous component  $(A_t)^{\vartheta-1}$ , which results from endogenous R&D and adoption. Labor productivity is directly empirically observable and model-implied labor productivity can be derived using equation (5) as:

$$\frac{Y_t}{L_t} = \underbrace{(A_t)^{\vartheta - 1}}_{\text{Endog. TFP TFP shock}} \underbrace{\theta_t}_{\text{TFP}} \left(\frac{U_t K_t}{L_t}\right)^{\alpha}$$

Consequently, model-implied labor productivity  $\frac{Y_t}{L_t}$  can be decomposed into total factor productivity as the product of endogenous and exogenous TFP component on the one hand and capital intensity, as measured as utilization-weighted capital per hours worked  $\frac{U_tK_t}{L_t}$  on the other hand.

#### 5.1 Empirical magnitude of endogenous TFP in the euro area

This section demonstrates that the model-implied pattern of TFP is in line with empirically observed productivity measures and that the endogenous component of total factor productivity, its endogenous component, as well as labor productivity. Both TFP and its endogenous component are identified from the model, while detrended labor productivity is empirically observed. Firstly, we find that labor productivity and total factor productivity display a high degree of comovement throughout the sample, suggesting that the implied evolution of TFP is in line with empirically observable labor productivity developments. Differences between labor productivity and total factor productivity and total factor productivity and total factor productivity result from fluctuations in the degree of utilization-adjusted capital intensity. Importantly, the endogenous component of TFP explains a large share of total factor productivity and both series closely comove throughout the sample period, suggesting that the endogenous component of TFP is of empirically relevant magnitude. In particular in the precrisis period, the initial upswing following the Great Recession until roughly 2012:2, as well as



Figure 8: Evolution of total factor productivity, endogenous TFP and labor productivity

towards the end of the sample period, the endogenous component of TFP nearly fully explains total factor productivity. Generally, the gap between total factor productivity and its respective endogenous component is attributed to exogenous TFP movements, i.e. the standard technology shock, which captures variation in total factor productivity not endogenously explicable by our model. Consequently, phases which are characterized by a relatively larger gap between TFP and its endogenous component are subject to a relatively higher importance of standard TFP shocks. That given, while endogenous TFP tracks total factor productivity closely throughout the sample, there is a more substantial unexplained wedge between TFP and its endogenous component during the Great Recession and also, albeit to a lesser extent, in the phase of roughly 2013:2 to 2016:4 than in the remaining periods. This suggests a more pronounced role of the TFP shock  $\theta_t$  during these phases. A potential explanation for this observation is related to changes in labor utilization over the business cycle as in particular during the Great Recession many euro area firms used labor hoarding, i.e. instead of laying off workers adjusted their labor input by means of reductions in hours and in labor utilization (see for instance ECB (2012)). While our employment measure (total hours of employees) captures reductions along the intensive margin, it does not capture the utilization of the respective hours worked which the model attributes to exogenous shifts in total factor productivity.

#### 5.2 Determinants and evolution of endogenous total factor productivity

We now turn to a more detailed analysis of the evolution and most relevant drivers of endogenous total factor productivity. As already discussed in the context of the variance decomposition (Ta-



Figure 9: Endogenous TFP and main shock contributions

ble 3), the liquidity demand shock constitutes the most important driver of euro area endogenous total factor productivity as it accounts for 71.6% of its variation. Figure 9 presents the evolution of endogenous TFP and the contribution of the liquidity demand shock and the shock to R&D efficiency respectively. We observe that the historical decomposition further confirms the key role of the liquidity demand shock in the evolution of endogenous TFP as both closely comove over the entire observation period. It is discernible that the decline in endogenous total factor productivity has already started in the early 2000s, which coincides with a persistent negative contribution of R&D efficiency over this period. Crucially, the speed of decline in endogenous TFP accelerated substantially during the Great Recession and can be explained by a strong negative contribution of the liquidity demand shock. Starting from 2013 the substantial TFP decline came to a halt and from 2016 onward TFP improvements are discernible. The TFP increase is explained by an alleviating negative contribution of the liquidity demand shock in line with overall improving general economic conditions in the context of the expansion. Declines in R&D efficiency, however, further constitute a drag on productivity.

#### 5.3 Stagnant innovation versus slowing technology adoption

Our model permits the decomposition of the euro area productivity slowdown into changes in R&D investment and technology adoption respectively. Figure 10 shows the evolution of the endogenous component of total factor productivity (green line), the stock of technologies (blue line) and the adoption rate (black line) over time. We observe that the main driving factor of the



Figure 10: Endogenous TFP, R&D and technology adoption

pre-crisis slowdown in euro area total factor productivity was the decline in the generation of new technologies through R&D activities, while the speed of technological diffusion as measured by the adoption rate was sound. The most relevant cause of the fall in productivity from the Great Recession onward, however, constituted a marked fall in technology adoption: The severity of the Great Recession substantially reduced firms' incentives to incorporate new technologies in the production process. Starting from about 2013, the degree of technological adoption has been increasing, in line with overall economic conditions and has represented the key driver of the starting recovery in total factor productivity.

## 6 Implications for euro area macroeconomic policy

We provided an analysis of euro area economic dynamics from the perspective of a DSGE model with endogenous total factor productivity mechanism in which total factor productivity evolves as the result of investment in R&D and technology adoption. Our results highlighted that when taking into account the endogeneity of total factor productivity, macroeconomic dynamics substantially differ in central aspects from standard DGSE models in which total factor productivity is exogenous. This section addresses the corresponding implications for macroeconomic policy in this context and in light of the identified main driving forces of the euro area productivity slowdown.

#### 6.1 Demand-side fluctuations matter for the evolution of the supply side

One of the main findings of this analysis, in line with the baseline model for the US case. is that demand-side shocks exert a significant influence on the evolution of productivity and that the latter displays a non-negligible degree of procylicality. This stands in sharp contrast to the underlying assumptions in standard macroeconomic frameworks, which assume purely supply-side determined productivity and hence rule out any direct effects from demand-side fluctuations to supply-side developments. This entails important policy implications. Firstly, our results suggest that negative demand effects in the recent euro area recession constitute an important contributor to the acceleration of the productivity slowdown in the euro area observed over the past decade. Moreover, our findings demonstrate that the feedback effects from the demand side to productivity represent an important source of business cycle amplification and thus an important explanation of the depth and persistence of the recent recession in the euro area. Lastly, we document a substantial degree of procyclicality of euro area total factor productivity. Hence, from the perspective of our analysis, when assessing the business cycle position and the degree of underutilization in the economy it is not sufficient to only take into consideration the extent of underutilization in production factors such as labor market slack and the deviation of the capital stock from its balanced growth path value but instead also the cyclical deviation of total factor productivity and its drivers through R&D and technology adoption from their respective long-run equilibrium levels. Standard output gap measures which only take into account the degree of underutilization in productive resources and do not factor in the corresponding cyclical shortfall in TFP may hence underestimate the degree of slack in the economy.

#### 6.2 Flattening of the traditional Phillips curve relationship

A further important policy implication is the flattening of the Phillips curve relationship over the business cycle under procyclical productivity dynamics. More specifically, inflation falls by less in downturns and increases less pronouncedly in upswings vis-à-vis standard macroeconomic workhorse models (see Figure 5). The underlying mechanism is that procyclical movements in TFP offset to a large extent the effect of economic slack on inflation. In an expansion, the diminishing degree of economic slack exerts an inflationary effect on prices which is, however, counteracted by a corresponding procyclical rise in productivity, reducing cost pressures. In a downturn, in turn, the general deflationary impact of the lower degree of capacity utilization is partly offset by a procyclical productivity drop which correspondingly exerts upward pressure on costs. In sum, the inherent endogenous TFP mechanism alleviates the inflation response over the business cycle. As a result, the traditional Phillips curve relationship, which predicts a positive relationship between the degree of slack in the economy and inflation is muted under procyclical productivity dynamics. This feature contributes in explaining the lack of pronounced responses of inflation during the Great Recession as well as in the subsequent expansion.

# 6.3 Reducing the depth of recessions: Alleviating the feedback to R&D and technology adoption

As the results in the previous sections demonstrated, the second-round effects of a recessionary shock on total factor productivity via a drop in R&D and technology adoption activity constitute the key mechanism, which induces deep and highly persistent recessions in this framework. Hence, policies which aim at reducing the feedback from overall economic conditions to productivity-enhancing investments in innovation and adoption and the corresponding downward spiral unfolding in the economy could for instance constitute apt tools in preventing the occurrence of deep recessions. Subsidies supporting R&D and technological diffusion can constitute suitable policy options to achieve this aim as in the presence of subsidies, an adverse shock hitting the economy could only exert limited effect on productivity. This holds true as the negative shock does not feed through to the R&D sector or firms' choice of adopting new technologies in production as the subsidies prevent technology-enhancing investments to fall below a certain threshold. To ensure the long-run sustainability of the subsidies to innovation and diffusion the subsidies could be designed counter-cyclically and only be paid in the event of a downturn, leaving the long-term output path unaltered.

As equations (9) and (15) demonstrate, R&D and technology adoption fall procyclically due to the concomitant drop in the expected value of an adopted or respectively unadopted technology as a result of decreasing expectations about firm profits. That given, maintaining innovation and adoption activity could be achieved by alleviating the costs of R&D and technology diffusion in the form of a lump-sum transfer. This could be realized by compensating firms' and innovators' wage costs and hence employment of high skilled labor, preventing the procyclical drop in productivity and a corresponding large-scale output drop.<sup>20</sup> As holds true for the proposed design of policies promoting productivity growth in this context, it is important to note that adequately designed policies should tackle both the procyclical decrease in R&D and technology adoption given the two-stage nature of technology growth. However, in the case that possible subsidies to technological progress are constrained, for instance in the case of limited fiscal space, subsidies to technology adoption should be prioritized as technological diffusion are subject to a higher degree of procyclicality. This suggests a higher effectiveness of subsidies allocated to this sector in preventing persistent downturns, rendering them a natural priority for policy making. This notwithstanding, the potential effect of subsidies to R&D should not be underestimated given their longer-term effect on also the potential for future technology adoption, as the maximum level of adopted technologies  $A_t$  constitutes straightforwardly a subset of and is hence restricted by the total stock of invented technologies  $Z_t$ .

<sup>&</sup>lt;sup>20</sup>The implicit assumption made here is that subsidy payments are conditional on actual increased innovation and adoption effort by firms. Put differently, firms cannot divert the subsidiy and use the transfer for other purposes.

# 7 Conclusion

We estimate a medium-scale DSGE model with an endogenous total factor productivity mechanism as proposed by Anzoategui, Comin, Gertler and Martinez (2018) on euro area data to analyze the mechanisms underlying the high degree of business cycle persistence in the context of the recent euro area recession, as well as the sources of the euro area productivity slowdown. We find that modeling the evolution of total factor productivity endogenously as a result of productivity-enhancing investments in R&D and technology adoption substantially increases euro area business cycle persistence vis-à-vis standard macroeconomic models with purely exogenous total factor productivity. Furthermore, we observe that the endogenous component of total factor productivity is empirically sizable and explains a high share of overall TFP in the euro area. Our results demonstrate a high degree of procyclicality of euro area total factor productivity. Regarding the evolution of euro area total factor productivity over time, we conclude that the slowdown in productivity has already set in in the early 2000s. During the Great Recession, however, we observe a marked acceleration of this decline. Concerning the main drivers of the slowdown in euro area productivity, we find that the fall in the efficiency of R&D investments in generating new innovations represents a main driving force in the pre-crisis phase, while starting from the Great Recession the liquidity demand shock, which favors safe asset holdings at the expense of consumption and the corresponding drop in firms' technology adoption constitutes the main contributing factor. This result crucially highlights the importance of demand-side shocks in explaining supply-side developments in the euro area. From 2015 onward, we document an increase in the endogenous TFP component, in line with overall improving economic conditions in the euro area. The diminishing efficiency of R&D investment in generating new innovations, however further constitute drags on euro area productivity. Finally, the endogenous productivity mechanism holds important implications for euro area inflation since its presence dampens the inflation response over the business cycle. More specifically, TFP falls when the economy is hit by a recessionary shock, raising production costs and muting the drop in inflation. This property contributes to the explanation of the negligible fall in inflation despite a severe output drop during the Great Recession. Likewise, productivity increases in an upswing in this setting and these procyclical productivity gains alleviate the rise in inflation in the event of an expansionary shock. Consequently, our results suggest that the procyclical increase of productivity has contributed to the sluggish response in euro area inflation in the current expansion.

# References

Aghion, P. and P. Howitt (1992): "A Model of Growth through Creative Destruction", Econometrica, Vol. 60 (2), pp. 323-351.

Anzoategui, D., D. Comin, M. Gertler and J. Martinez (2018): "Endogenous Technology Adoption and R&D as Sources of Business Cycle Persistence", American Economic Journal: Macroeconomics (forthcoming).

Benigno, G. and L. Fornaro (2017): "Stagnation Traps", Review of Economic Studies, Vol. 85 (3), pp. 1425-1470.

Bianchi, F., H. Kung and G. Morales (2019): "Growth, Slowdowns, and Recoveries", Journal of Monetary Economics, Vol. 101, pp. 47-63.

Bosworth, D. L. (1978): "The Rate of Obsolescence of Technical Knowledge - A Note," Journal of Industrial Economics, Vol. 26(3), pp. 273-279.

Broda, C. and D. E. Weinstein (2006): "Globalization and the Gains From Variety", Quarterly Journal of Economics, Vol. 121(2), pp. 541-585.

Caballero, R. J. and A. B. Jaffe (1993): "How High are the Giants' Shoulders: An Empirical Assessment of Knowledge Spillovers and Creative Destruction in a Model of Economic Growth," in NBER Macroeconomics Annual 1993, Volume 8, National Bureau of Economic Research, Inc, NBER Chapters, 15-86.

Christiano, L. J., M. Eichenbaum, and C. L. Evans (2005): "Nominal Rigidities and the Dynamic Effects of a Shock to Monetary Policy," Journal of Political Economy, Vol. 113(1), pp. 1-45.

Coibion, O., Y. Gorodnichenko and M. Ulate (2017): "The Cyclical Sensitivity in Estimates of Potential Output", NBER Working Papers 23580.

Comin, D. and M. Gertler (2006): "Medium-Term Business Cycles", American Economic Review, Vol. 96(3), pp. 523-551.

Comin, D. and B. Hobijn (2010): "An Exploration of Technology Diffusion," American Economic Review, Vol. 100(5), pp. 2031-59.

Comin, D. and M. Mestieri (2018): "If Technology Has Arrived Everywhere, Why has Income Diverged?", American Economic Journal: Macroeconomics, Vol. 10 (3), pp. 137-178.

Comin, D., M. Gertler and A. M. Santacreu (2018): "Technology Innovation and Diffusion as Sources of Output and Asset Price Fluctuations", Working Paper.

Cox, M. W. and R. Alm (1996): "The economy at light speed: technology and growth in the information age and beyond," Federal Reserve Bank of Dallas, Annual Report, pp. 2-17.

Eggertsson, G. B., N.R. Mehrotra and J. Robbins (2019): "A Model of Secular Stagnation: Theory and Quantitative Evaluation", American Economic Journal: Macroeconomics, Vol. 11(1), pp. 1-48.

Eggertsson, G. B., N.R. Mehrotra, S. Singh and L. Summers (2016): "A Contagious Malady? Open Economy Dimensions of Secular Stagnation", IMF Economic Review, Vol. 64(4), pp. 581-634.

ECB (2012): "Euro Area Labour Markets and the Crisis", ECB Occasional Paper Series, No. 138.

Fisher, J. D. (2015): "On the Structural Interpretation of the Smets-Wouters 'Risk Premium' Shock," Journal of Money, Credit and Banking, Vol. 47(2-3), pp. 511-516.

Gordon, R. J. (2015): "Secular Stagnation: A Supply-Side View", American Economic Review, Vol. 105 (5), pp. 54-59.

Greenwood, J., Hercowitz, Z. and G. W. Huffman (1988): "Investment, Capacity Utilization, and the Real Business Cycle," The American Economic Review, Vol. 78(3), pp. 402-417.

Griliches, Z. (1990): "Patent Statistics as Economic Indicators: A Survey", Journal of Economic Literature, Vol. 28 (4), pp. 1661-1707.

Grossman, G. and E. Helpman (1991): "Quality Ladders and Product Cycles", Quarterly Journal of Economics, Vol. 106(2), pp. 557-586.

Jaimovich, N. (2007): "Firm Dynamics and Markup Variations: Implications for Sunspot Equilibria and Endogenous Economic Fluctuations", Journal of Economic Theory, Vol.137(1), pp. 300-325.

Kortela, T. (2016): "A Shadow Rate Model with Time Varying Lower Bound of Interest Rates", Bank of Finland Research Discussion Paper.

Krishnamurthy, A. and A. Vissing-Jorgensen (2012): "The Aggregate Demand for Treasury Debt," Journal of Political Economy, Vol. 120(2), pp. 233-267.

Kung, H. and L. Schmid (2015): "Innovation, Growth and Asset Prices", Journal of Finance, Vol. 70(3), pp. 1001-1037.

Romer, P. M. (1990): "Endogenous Technological Change", Journal of Political Economy, Vol. 98(5), pp. 71-102.

Smets, F. and R. Wouters (2007): "Shocks and Frictions in US Business Cycles: A Bayesian DSGE Approach", American Economic Review, Vol. 97(3), pp. 586-606.

Wu, J. C. and F. D. Xia (2017): "Time Varying Lower Bound of Interest Rates in Europe", Working paper.

# A Appendix

#### A.1 Data

The data used in the estimation are accessible from the Eurostat Database (ec.europa.eu/ eurostat/de/data/database) and the ECB Statistical Data Warehouse (sdw.ecb.europa.eu). The model is estimated using series on real GDP, investment, consumption, hours worked, compensation of employees, total population and nominal interest rates. All series are denoted in terms of the euro area fixed composition (EA-12) and are - when applicable - seasonally- and calendar-adjusted.

Real GDP is denoted in terms of Million euros and chain linked volumes with reference year 2010. The series on consumption measures household and NPISH final consumption expenditure in Mio. Euros and current prices. Our measure of investment is gross fixed capital formation denoted in current prices. Hours worked correspond to total hours worked expressed in 1000 hours using the domestic employment concept. Compensation refers to compensation of employees (denoted in Mio. Euro and current prices). Population corresponds to the total population in the EA-12 countries and the GDP deflator refers to the price index. Lastly, the series on R&D investment is annual<sup>21</sup> and refers to business expenditure on research and development, where the business sector corresponds to the source of funds. We use as a proxy for the nominal interest rate the 3-months Euribor and the 3-months shadow rates as estimated by Kortela (2016)<sup>22</sup> in the ZLB episode.<sup>23</sup> Based on these data, we construct the series used in estimation as follows, where  $\Delta$  denotes the temporal difference operator:

- Output growth =  $100 \times \Delta \operatorname{LN}\left(\frac{\operatorname{Real GDP} \times 1000}{\operatorname{Population}}\right)$
- Consumption growth =  $100 \times \Delta \operatorname{LN}\left(\frac{\frac{\operatorname{Consumption} \times 1000}{\operatorname{GDP deflator}}}{\operatorname{Population}}\right)$
- Investment growth =  $100 \times \Delta \operatorname{LN}\left(\frac{\frac{\operatorname{Investment} \times 1000}{\operatorname{GDP} \operatorname{deflator}}}{\operatorname{Population}}\right)$
- Real wage growth =  $100 \times \Delta \operatorname{LN}\left(\frac{\frac{\operatorname{Compensation} \times 1000}{\operatorname{Hours}}}{\operatorname{GDP deflator}}\right)$
- Growth rate of hours worked =  $100 \times \Delta \operatorname{LN}\left(\frac{\operatorname{Hours}}{\operatorname{Population}}\right)$
- Inflation =  $100 \times \Delta$  LN (GDP deflator)

<sup>&</sup>lt;sup>21</sup>We deal with the mixed-frequency issue in the data by means of a Kalman-filter approach.

 $<sup>^{22}</sup>$ The shadow rates estimated by Wu and Xia (2017) are used as a robustness check.

<sup>&</sup>lt;sup>23</sup>We take a shadow rate approach as the shadow rate provides a more comprehensive measure of the monetary policy stance, including the role of non-standard monetary policy measures. As a robustness check, we use an alternative model specification which imposes the ZLB constraint and we extract smoothed variables and corresponding shock contributions using the method developed by Anzoategui (2017). The corresponding results are highly similar to the findings in our baseline model, where the key difference constitutes the somewhat higher role attributed to the liquidity demand shock in explaining economic fluctuations in the nonlinear model.

- Nominal interest rate  $=\frac{1}{4} \times \text{Euribor}$
- Growth rate of R&D investment =  $100 \times \Delta \operatorname{LN}\left(\frac{\frac{\operatorname{R&D investment \times 1000}}{\operatorname{GDP deflator}}}{\operatorname{Population}}\right).$

# Bank of Finland Research Discussion Papers 2019

ISSN 1456-6184, online

1/2019	Marco Gallegati – Federico Giri – Michele Fratianni Money growth and inflation: International historical evidence on high inflation episodes for developed countries ISBN 978-952-323-257-0, online
2/2019	Francesco D'Acunto – Daniel Hoang – Maritta Paloviita – Michael Weber <b>IQ, Expectations, and Choice</b> ISBN 978-952-323-258-7, online
3/2019	Orkun Saka Domestic banks as lightning rods? Home bias and information during Eurozone crisis ISBN 978-952-323-260-0, online
4/2019	Saara Tuuli <b>Model-based regulation and firms' access to finance</b> ISBN 978-952-323-261-7, online
5/2019	Thomas A. Lubik – Christian Matthes – Fabio Verona Assessing U.S. aggregate fluctuations across time and frequencies ISBN 978-952-323-263-1, online
6/2019	Karlo Kauko – Eero Tölö <b>On the long-run calibration of the Credit-to-GDP gap as a banking crisis predictor</b> ISBN 978-952-323-264-8, online
7/2019	Olli-Matti Laine <b>The effect of TLTRO-II on bank lending</b> ISBN 978-952-323-269-3, online
8/2019	Michaela Schmöller Stagnant wages, sectoral misallocation and slowing productivity growth ISBN 978-952-323-273-0, online
9/2019	Adam Gulan – Markus Haavio – Juha Kilponen Can large trade shocks cause crises? The case of the Finnish-Soviet trade collapse ISBN 978-952-323-274-7, online
10/2019	Ewa Stanisławska – Maritta Paloviita – Tomasz Łyziak Assessing reliability of aggregated inflation views in the European Commission consumer survey ISBN 978-952-323-277-8, online
11/2019	Patrick M. Crowley – David Hudgins U.S. Macroeconomic Policy Evaluation in an Open Economy Context using Wavelet Decomposed Optimal Control Methods ISBN 978-952-323-279-2, online
12/2019	Luís Aguiar-Conraria – Manuel M. F. Martins – Maria Joana Soares The Phillips Curve at 60: time for time and frequency ISBN 978-952-323-280-8, online
13/2019	Gene Ambrocio – Iftekhar Hasan Friends for the benefits: The effect of political ties on sovereign borrowing conditions ISBN 978-952-323-282-2, online
14/2019	Eero Tölö <b>Predicting systemic financial crises with recurrent neural networks</b> ISBN 978-952-323-287-7, online

- 15/2019 Tuomas Takalo Switching costs in the Finnish retail deposit market ISBN 978-952-323-289-1, online
- 16/2019 Sami Alpanda Eleonora Granziera and Sarah Zubairy State dependence of monetary policy across business, credit and interest rate cycles ISBN 978-952-323-290-7, online
- 17/2019 Gene Ambrocio Measuring household uncertainty in EU countries ISBN 978-952-323-291-4, online
- 18/2019 Delong Li Lei Lu Congming Mu Jinqiang Yang Biased beliefs, costly external finance, and firm behavior: A Unified theory ISBN 978-952-323-293-8, online
- 19/2019 Christopher G. Gibbs Nigel McClung Does my model predict a forward guidance puzzle ISBN 978-952-323-294-5, online
- 20/2019 Qazi Haque Nicolas Groshenny Mark Weder Do we really know that U.S. monetary policy was destabilizing in the 1970s? ISBN 978-952-323-295-2, online
- 21/2019 Michaela Schmöller Martin Spitzer Endogenous TFP, business cycle persistence and the productivity slowdown ISBN 978-952-323-296-9, online